Author's response to reviews

Title: Prevalence of Diabetic Retinopathy in Tehran Province: A Population-Based Study

Authors:

Nasrin Rafati (labbafi@hotmail.com)
Mohammad Hossein Dehghan (Mhdehghan5@hotmail.com)
Marzieh Katibeh (mdkatibeh@yahoo.com)
Farid Zayeri (labbafi@hotmail.com)
Mehdi Yaseri (m.yaseri@yahoo.com)
Mohammad Ali Javadi (labbafi@hotmail.com)
Mojtaba Sehat (om_sehat@yahoo.com)
Hamid Ahmadieh (hahmadieh@hotmail.com)

Version: 2 Date: 25 January 2009

Author's response to reviews: see over
Dear all

The attached file is our manuscript entitled “Prevalence of Diabetic Retinopathy in Tehran Province: A Population-Based Study”. This is an original paper that has not been published in any languages elsewhere. In addition, this manuscript, or substantial parts of it, is not submitted and under consideration by any other journal. All authors have read and agreed to its content and it should be mention that the authors have no financial interest in this study. We will be very grateful if you consider it for peer review evaluation and publication in your valuable journal.

Here, we should add that the present manuscript is based on the results of a time-consuming and extensive team work in different areas of Tehran province, Iran. Our executive and scientific team consisted of ophthalmologists (Dr Rafati, Dr Dehghan and Professor Ahmadieh), epidemiologists (Dr Sehat and Dr Katibeh) and biostatisticians (Dr Zayeri and Mr Yaseri).

If there is anything else we should do, please do not hesitate to ask. We are eagerly looking forward to hearing from you.

Marzieh Katibeh, MD, MPH,
Assistant Professor of Community Medicine,
Ophthalmic Research Center,
Labbafinejad Medical Center,
Shahid Beheshti University, M.C. Tehran-Iran
Tel: +98-21-22585952, Fax: +98-21-22587317
E-mail: mdkatibeh@yahoo.com